Author's response to reviews

Title: Mental health first aid training of the public in a rural area: a cluster randomized trial [ISRCTN53887541]

Authors:

Anthony F Jorm (anthony.jorm@anu.edu.au)
Betty A Kitchener (betty.kitchener@anu.edu.au)
Richard O'Kearney (richard.okearney@anu.edu.au)
Keith BG Dear (keith.dear@anu.edu.au)

Version: 2 Date: 20 October 2004

Author's response to reviews:

The reviewer recommended that the paper be accepted without changes. As requested by the editorial staff, the following changes were made:
1. Reference 5 by Caldwell et al has been updated.
2. The description of additional files has been removed.